

# **Framing Employer Branding**

# An explorative case study of legitimacy within the public sector context

Author: Josefine Koskinen

Essay/Thesis:	30 hp
Program and/or course:	Strategic HRM and Labour Relations
Level:	Master Thesis
Semester/year:	Spring 2015
Supervisor:	Bertil Rolandsson
Examiner:	Lars Hansen

# Abstract

**Title:** Framing Employer Branding - An explorative case study of legitimacy within the public sector context.

Author: Josefine Koskinen

**Background:** Increasingly, organizations from the public sector embrace employer branding as a strategy to better attract and retain qualified employees. Current research shows a lack of research regarding how public sector organizations' work with employer branding, where the mission of the organization is not to conquer the global market but rather to provide welfare and serve the common good for all citizens. Hence, public sector organizations cannot just compete by being attractive as an employer; they also have to present their assignment as something attractive for the applicants. The purpose of this study is to explore how one municipality within the public sector in Sweden frame employer branding to make the alignment between the strategy and the organizations' overall goals and mission legitimate.

**Method:** The study uses a qualitative research strategy with the approach of a single case study of one municipality within the public sector in Sweden. 15 semi-structured interviews were conducted with HR specialists, and managers from different departments within the Municipality of Gothenburg.

**Results:** The findings show that employer branding is used as strategy to help the organization deal with issues of legitimacy. A dilemma occurs to exist as the organizations' aim to practice employer branding as an act of balance between internal unity and external sensitivity towards diversity. The dilemma creates uncertainty among employees' that show a tendency to use mimetic framing as a strategy since they experience the situation to be ambiguous. The results from this study show an importance for HR to take on a leadership role and function as support, and make sure to educate employees about this dilemma. Therefore public sector organizations need to consider the importance to be transparent and communicate about this dilemma, and make sure that employees' are aware of different opportunities that exist.

Key words: Employer branding, legitimacy, common good for all, public sector, frame theory

# Acknowledgement

The making of this thesis has truly been an exiting journey! I am extremely thankful to all the people that I have around me that have been giving me any type of support in the making of this report.

I would like to express my deepest gratitude to my family and friends; thank you for encouraging and inspiring me. I am more than thankful that I have such amazing people around me. This thesis would not have been possible without you.

Secondly, I would also like to express gratitude to my supervisor Bertil Rolandsson. You have helped me through this progress of the thesis with huge dedication. I am thankful for all the guidance and support you been giving me. Thank you!

Additionally, I would also like to give special thanks to all the participants in this study, for giving me some of your time and for sharing valuable thoughts.

Josefine Koskinen Stockholm 4 juni 2015

# Table of content

1. Introduction	1
1.1 The need for qualified human capital	
1.2 Problem statement	2
1.3 Purpose and Research Questions	4
1.4 Background	4
1.5 Outline of the thesis	
2. Previous Research	6
2.1 Employer perspective	6
2.2 Brand communication	8
2.3 Employee perspective	9
3. Theoretical framework	10
3.1 Institutional theory and Legitimacy	
3.2 Frame theory	
3.2.1 Different types of framing	
3.3 Goal Framing Theory	
4. Methodology:	14
4. 1 Research strategy	
4.2 Research approach	
4.3 Case context	
4.4 Research design	
4.5 Data collection	
4.5.1 Sampling	
4.5.2 Interviews	
4.6 Data Analysis	
4.6.1 Trustworthiness	
4.7 Ethical considerations	
4.8 Limitations of the research method	
5. Results	20
5.1 Organizing coherent employer branding	
5.1.1 Employer branding activities	
5.1.2 Match-Making	
5.1.3 The water drop effect	
5.2 Organizing employer branding sensitive to diversity	
5.2.1 Humanistic perspective	
5.2.2 Frustration over the abstract and visional	
5.2.3 Ambiguity	27
6. Discussion and analysis	
6.1 Organizing coherent employer branding	

6.2 Organizing employer branding sensitive towards diversity	31
6.3 Organizing coherent employer branding sensitive towards diversity – A dilemma?	33
7. Conclusion	35
8. Final concluding remarks	37
8.1 Suggestions for future research	37
9. References:	38
10. Appendix	44
Appendix 1 – Interview guide	
Appendix 2 - Letter of consent	45

### 1. Introduction

#### 1.1 The need for qualified human capital

The war for talents is claimed to be a strategic business challenge that companies need to battle as their viability depends on it (Sivertzen, Nilsen Volafsen, 2013). Without the capability to recruit and retain qualified human capital, companies are argued to face difficulties since most companies build their operations around competencies (Lindmark& Önnevik, 2011). Employer branding has frequently been used within private sector organizations as a strategy to attract and retain qualified human capital. Increasingly, organizations from the public sector are applying strategies to manage their brand more efficiently, and to better attract employees (Parmar, 2014; Dahlqvist& Melin, 2010). If public sector organizations cannot attract and retain employees with the right competencies there will be societal implications as they may not be able to provide the same quality on their services. Hence, it is not only the organizations' viability that depend on the public sector's capability to attract and retain human capital, the whole society will be affected as public sector organizations' manage services that are used throughout citizens' lives. Employer branding is a concept that is used as a strategy for how to attract and retain talent, and it is said to be focusing on "strategic alignment of human capital with organizational goals" (Backhaus& Tikoo, 2004:511). Minchington (2010) argues that employer branding needs to be integrated in the company's overall strategy-work in order to become successful. However, how is the strategic alignment of human capital done in public sector organizations in Sweden, where they have to consider other types of goals? How do public sector organizations work with employer branding in order to make the alignment between their employer branding strategy and their overall mission and goals legitimate? This will be the topic of this thesis that will focus on the public sector within the Swedish context.

Public sector organizations cannot just compete by being attractive as an employer; they also have to present their assignment as something attractive for the applicants. Public sector organizations are predominantly politically driven, and work under guidelines from determined political goals, which are common goals that should be assessable for every citizen. Political goals aim at supporting collective interest with the aim of achieving common good for all citizens within the given society (Johnson& Scholes, 2001). The context of public sector

organizations therefore differ somewhat from the private sector context as the mission is to achieve common good for all citizens, thereof the interest to explore some of the challenges and conditions that come into play regarding how public sector organizations' frame employer branding. How do public sector organizations' work with employer branding, where the goal of business is not to conquer the global labor market, but rather to fulfill their mission to serve high quality service for the common good of all citizens (Dahlqvist& Melin, 2010). The legitimacy of the organization depends very much upon how public sector organizations are framed within society. People use frames as tools to organize and describe events in their surroundings, and through the practice of framing people make sense of and process information (Boundarouk, Looise& Lemsink, 2009).

This report will explore the case of the Municipality of Gothenburg to see how they practice employer branding in order to endorse their legitimacy. The Municipality of Gothenburg has around 50 000 employees working within the organization that is built upon different types of business operations (Göteborgs stad:A, n.d). The organization has ten city district administrations that aim to manage and provide public services to the population within their own district. Additionally, there are around 20 different specialized departments and multiple state enterprises, which focus on more specialized goals, such as health services, education and social services (Göteborgs stad:B, n.d). The Municipality of Gothenburg needs to manage to attract employees with the right competences in order to keep the different businesses within the organization running.

#### **1.2 Problem statement**

There are several challenges to the public sector that are apposed to efficiently manage employer branding aligned with the public goals. For instance, Dahlqvist and Melin (2010) claim that large parts of the public sector organizations have image issues that they are battling with. In today's, technology driven (Chhabra& Sharma, 2014), demographic, attitudinal-shifting, competitive labor market (Moroko& Uncles, 2008; Edwards, 2005) it is crucial to maintain good reputation and build an attractive image to be able to attract qualified employees that can contribute, and enhance companies competitive advantage (Backhaus& Tikoo, 2004). Ambler and Barrow (1996) assert that reputation is an important mechanism in employer branding, as the brand bring about associations, ideas, images and feelings about how it is to be working at that particular

company. Research has shown that potential recruits tend to view companies with good reputation as more attractive employers than companies with less great reputation (Edwards, 2009). Reputation is an important key for attracting qualified employees that can assist in the work of achieving overall strategic goals of the organization. Hylen (2011) claims that talents that make career within municipalities need to act as ambassadors, and sources of inspiration, in order to make municipalities attractive employers in the eyes of potential applicants. 'Sveriges Kommuner och Landsting' (SKL) (2006) claims that many people holds the view of employment within the public sector to be dull and non-attractive, which makes reputation a huge challenge in public sector organizations' employer branding work. Municipalities work to provide highquality services to serve the common good for all citizens (Government offices of Sweden, 2011) and SKL (2010) claims that it is crucial for Swedish municipalities to attract the right employees in order to continue deliver quality-service to all citizens. Without employees that acquire the right competencies, the public sector will fail to deliver good quality on their services, which will have negative impacts on society. Municipalities and county councils work to uphold good quality of the Swedish welfare system, and they aim to provide common good for all citizens (ibid). Education, from pre-school to high school level, eldercare, and hospitality are some of the services that public sector organizations provide (Dahlqvist& Melin, 2010). The societal implications will be devastating if public sector organizations' fail to attract and retain qualified human capital, and the whole idea of the common good for all citizens might come to an end.

Employer branding is arguably a strategic issue for HRM as the company's viability depends on their capability to manage human capital in successful manners (Backhaus& Tikoo, 2004). Christiaans (2013) asserts that employer branding can be put in the category of Strategic HR management, since employer branding focuses on the overall strategic goals of the organization. However, there is not much research done that takes the HR function into consideration in relation to employer branding, which leaves the context relatively unexplored (Parmar, 2014). Neither does current research of employer branding (EB) reflect on the evident differences between the private- and public sector (Dahlqvist& Melin, 2010). The application of employer branding similar to the private sector, may not be the best option for how to practice employer branding within the public sector. Literature of employer branding tends to be very descriptive and normative with little empirical research of different practices and what actually is done in

terms of employer branding (Edwards, 2009). There is a need for more empirical research of how public organizations practice employer branding to further explore what the concept entails in the Swedish local public administration context.

#### **1.3 Purpose and Research Questions**

The purpose of this study is to explore how the public sector in Sweden practice employer branding and how they work in order to make the alignment between employer branding and the organization's overall goals and mission legitimate. Further, the study aims to explore how public sector organizations in a Swedish work to keep their legitimacy by investigating how employees' frame employer branding. This in order to gain further knowledge and contribute with research to amplify the relatively unexplored area of employer branding within public sector organizations. Following research questions have been developed:

**RQ1:** How do a public sector organization in the Swedish context practice employer branding?

**RQ2:** *How is employer branding aligned with the overall mission and goals of a public sector organization in Sweden?* 

#### 1.4 Background

These questions will be answered by looking at the specific case of a Swedish Municipality within the public sector, more specifically, the Municipality of Gothenburg. More than a million people work within the public sector/welfare system, and within a span of ten years, about one third of these employees will retire (SKL, 2015: Dyhre& Parment, 2013; Parment et al, 2009). This creates and urgency for organizations to recruit employees to different positions within the public sector. If nothing gets done regarding this matter, the welfare system is going to need over a half million new employees until year 2023 (SKL, 2015). It was in 2008 that the Municipality of Gothenburg acknowledged that they were about to face difficult times regarding their recruitment of qualified employees with the right competences. A project for employer branding was created, which they called "Attractive Employer" (translated from Göteborgs stad, 2011), in which they discussed challenges, and came up with strategies for how to deal with these. About the same time, the Swedish Municipalities and County Councils organization, which in Sweden is shortly referred to as SKL (Svenska Kommuner och Landsting), recognized that there were

existing recruitment issues, and stated that the problems related to recruitment had to become key priorities, which all organizations within the public sector in Sweden should start working more with. In 2014, SKL launched a project called 'Sweden's Most Important Jobs" (Sveriges Viktigaste Jobb, 2014) with the aim to spread knowledge about work within Swedish municipalities (ibid). An additional aim was to try to help change the picture of municipalities, this in order to increase positive associations, and ideally contribute to spread positive reputation about Swedish municipalities as attractive employers (ibid). The project of "Sweden's most important jobs" can be placed on a more national level, with the aim to spread attention to the recruitment-issues that all municipalities within Sweden share (SKL, 2010). This study's case of the Municipality of Gothenburg can be place within SKL's project and affirmed key priorities. This report will investigate, and explore how the Municipality of Gothenburg practice employer branding in order to enhance their employer attractiveness. The research area of this report goes along with the concept of New Public Management, which discusses the transfer of ideas from the private to public sector as well how the governance of public sector should look like (Røvik, 2008). There is much knowledge and research of employer branding within the private sector, but little about employer branding within the public sector. Employer branding is a concept with origins in the private sector, but it has started to spread across different business sectors. However, what type of trends can we expect when we talk about employer branding within the public sector?

#### 1.5 Outline of the thesis

Chapter one gives an introduction to employer branding and some of the related challenges that public sector organizations might confront. The chapter aims to give background information to give significance to the purpose and objectives of the study. Chapter two reveals important aspects found through extensive literature reviews of previous research, and some of the theories that been used within the field are presented. Chapter three gives an introduction and explains institutional theory and concepts within the frame theory, which will serve as the theoretical foundation of this study. Chapter four provides a thorough description of the study's methodology. The results of the empirical findings are presented in themes that evolved, these can be found in chapter five. Furthermore, the sixth chapter includes discussion and analysis of the research findings. The conclusion in chapter seven summarizes the important aspects of the

study, and argues for the significance of the thesis. Additionally, suggestions for future research are presented in the end of the final remarks section.

#### 2. Previous Research

This chapter gives an introduction to some of the main aspects found within employer branding literature with the aim to provide the reader with some further understanding of the field.

#### 2.1 Employer perspective

Within management research it is argued that employer branding has turned into a prominent strategic tool that can be used to attract and retain talent (Chhabra& Sharma, 2012). Studies have shown that labor market shortages and global competition are some of the underlying factors to organizations' interest in employer branding. Another factor that might influence is the fact that employees nowadays seem more willing to change jobs more often (Rampl& Kenning, 2014). For instance, Dyhre and Parment (2013) argue that companies need to start to communicate about different career paths available within their organization, since research shows that an employer get more attractive in the eyes of prospective employees if career paths within the company are easily identifiable (ibid). Backhaus and Tikoo (2004:511) argue that employer branding is an exercise of promoting career paths and individual development within the organization, and assert that without "career advancement, individuals lose their competitive edge, and so does the firm". Dahlqvist and Melin (2010) claim that a large amount of municipalities in Sweden have started to realize the importance of building strong brands as essential for their sustained competitive advantage. Many organizations have started working on different types of brand-projects. However, Dahlqvist and Melin (2010) assert that many of these projects unfortunately end up in piled documents, filled with wise words about how to become an attractive employers, but without any practical actions taking place.

Employer branding is assumed to be build upon the beliefs that human capital can bring value and provide loads of benefits to the firm, and that companies can enhance their performance by managing employees efficiently (Moroko& Uncles, 2008). For instance, research shows that employer branding can positively affect the organizational culture, as well as the productivity of the organization (Parmar, 2014). Backhaus and Tikoo (2004) present research results that show how organizations noticed that effective employer branding lead to competitive advantage, and helped employees internalize the values of the company. Successful employer branding is said to be able to attract human capital that best fits the company, and that can contribute to the strategic aims, and the bottom line of organizations (Wallace, Lings, Cameron& Sheldon, n.d). There is research that has shown that employer branding can help reduce costs and lead the way towards increased financial results (Barrow& Mosley, 2005)

Employer branding has become an important tool in HR practitioner's toolkit, and Edwards (2005:269) claims that, "The notion of employer branding pulls the HR function into the strategic engine room of the organization, encouraging HR professionals to become Ulrich's (1997) strategic partners". However, current research reveals issues and discussions regarding questions of whom stands, or should stand, responsible for employer branding within organizations. The discussions seem to be around HR and marketing, for instance, Wallace et al., (n.d) claim that there is a need for organizations to strategically align the HR functions with the marketing and communication functions in order to better attract and retain talent. Foster et al., (2010) argue that HR and marketing practices need to be more coordinated, and Backhaus and Tikoo (2004) assert that strategies within marketing and HR should be congregated. It is argued that both sides can learn from each other and benefit from collaborating, which possibly could lead to a more consistent and successful employer brand (Barrow& Mosley, 2005). Dahlqvist and Melin (2010) claim that a successful brand is defined from the outside of the organization to the inside of the organization, but built from the inside out. Backhaus and Tikoo (2004) explain that practitioners within the HRM literature describe employer branding as a three-step process: first, the company needs to develop its Employee Value Proposition (EVP), which is the existing package of benefits that the company has and can offer to prospective and current employees (Moroko& Uncles, 2008). The second step is external marketing that aims to attract ideal applicants, and the third step is internal marketing of the employer brand with the goal to develop a workforce that is committed to the goals and values of the organization (Backhaus& Tikoo, 2004). A second approach within research holds the perspective that companies should take on an inside-out approach with the assumption that happy and satisfied employees will

function as proactive brand ambassadors, and contribute to a positive reputation of the company (Van Mossevelde, 2012).

#### 2.2 Brand communication

Previous research holds the perspective of employer branding to be a concept that can function as a strategy to communicate about the company's values, and additionally help to make sure that employees fit with the organization's culture (Parmar, 2014). Employer branding is said to impact the culture of the organization as well as its identity, which is claimed to contribute to employer brand loyalty (Backhaus& Tikoo, 2004). Backhaus and Tikoo (2004) describe brand loyalty as the commitment that employees make to their employer. Employer branding is claimed to allow the company to attract employees that hold similar values as the organization and is argued that companies have started to realize the importance of recruiting employees that hold values that fit with the organization's values (Foster et al, 2010).

Although most researchers argue that employer branding aims to communicate to both prospective and existing employees, Foster et al., (2010) assert that most current research focuses on effective value-matching and communication at a person-organization level during the recruitment process. Dyhre and Parment (2013) claim that communication is important, and that organizations should work with internal communication in order to create a strategy for how to communicate inside the organization in order to create a coherent workforce in which all employees are aware of what is going on. The value of communication has been recognized as an important tool to create efficiency, and to reach the goals of the organization (Dahlqvist& Melin, 2010). They claim that one criterion for successfully building a strong brand is transparency with a coherent connection between the mission and what the company offers (ibid). A lack of transparency and consistency between what the company offers, and the mission, is argued to be common among municipalities branding efforts (ibid). It is important to be transparent both internally and externally, when it comes to what goes on in the employer branding process (Dahlqvist& Melin, 2010).

In order for public sector organizations to gain acceptance, and keep their legitimacy, they need to show citizens within society that they deliver high-quality service for the common good, and provide measurable results (ibid). Dahlqvist and Melin (2010) argue that the picture of public

sector organizations is affected by values that the organization communicates, which are assumed to be built upon values, experiences, reputation and the image media paints of the organization. The value of a brand is said to lie in the eyes of the observer (ibid), and Edwards (2005) claims that there have been an increase in job applicants that want to work for a company that stands for particular values, ideally similar to their own. Edwards and Kelan (2011) discuss employer branding in relation to diversity as the authors' believe diversity and employer branding can be each other's opposite. Diversity is described as the process of bringing out the differences between employees (ibid). They describe branding as something that aims towards creating an aligned workforce and coherent employment brand, and argue that the aim can lead to a homogenization of the workforce, creating issues of diversity (ibid).

#### 2.3 Employee perspective

The perspective of the psychological contract, which builds on work from social exchange theory, has commonly been used within employer branding research. The messages that are communicated by companies to attract prospective employees can be interpreted as promises that the company makes to the employees (Christians, 2013). Employer branding is said to involve strategies with the aim to develop a consistent positive image that will increase the employer attractiveness to help organizations' attract applicants, ideally from the desired target group (Ambler& Barrow; 1996). The concept of employer attractiveness is usually taken into consideration when describing employer branding, and is described as the mental image of benefits that prospective employees hold that they would gain when working for a particular employer (ibid). Christiaans (2013) claims that it is important the benefits that are communicated paint up an accurate picture in order to avoid developing unrealistic expectation among prospective employees. The psychological contract is about fairness perceptions, and is describes as a reciprocal exchange agreement between the employee and the employer (Edwards, 2009). Backhaus and Tikoo (2004) argue that employer branding is the start of the formation of a psychological contract. Christiaans (2013) claims that literature on the psychological contract perspective can be used to gain better understand of the crucial role of realistic employment images. Welinder, Araujo and Lynn (2011) describe employer branding as the process of creating an identity, and communicating that identity to ensure that it remains authentic.

An individual who identifies with the organization is argued to be likely to accept the goals and values of the organization, and is according to Edwards (2005) very likely to stay within the organization. When employees' care about the brand they are usually committed and work hard for the company they work for. Barrow and Mosley (2005) argue that it is the extra effort that makes the difference in today's competitive market. A theoretical perspective that has commonly been used within employer branding research is the person-organization fit perspective, which has been used to help explain the importance of congruence of values between employees and employers (Christiaans, 2013). It is argued that research of person-organization fit indicate that prospective employees compare the image of the employer brand to their own needs, personalities and values. An individual that has matching values with the organizational values is more likely to be attracted to that particular organization (Backhaus& Tikoo, 2004). An additional theory that commonly has been used within employer branding research is the social identity theory (ibid). According to the social-identity theory, employees develop their social identity and self-esteem from their organizational membership (Christiaans, 2013), and in addition, the reputation of the social group can contribute to peoples' self-concept (Backhaus& Tikoo, 2004).

#### 3. Theoretical framework

Evoking back to the purpose, this study explores how public sector organizations in Sweden practice employer branding with the aim to see how they work in order to make the alignment between employer branding and the organization's overall goal and mission legitimate. This study will be conducted by investigating how employees in the case of the Municipality of Gothenburg frame employer branding. Employees use various frames as a way to organize experiences and to make sense of occurring events in their environment. The purpose of this study will be addressed theoretically by using concepts within institutional theory that discuss the concept of legitimacy. Additionally, frame theory is used in order to be able to investigate how employees actively make sense of employer branding, and legitimize their actions.

#### 3.1 Institutional theory and Legitimacy

The institutional theory builds on the assumption that organizations are embedded in a wider institutional environment in which the organizational practices are the result of rules and structures that exist within the larger institutional environment (Paauwe& Boselie, 2003; Najeeb, 2014). Institutional theory asserts that legitimacy has a fundamental role in social interactions as it is defined as generalized perceptions of social acceptance (Vergne, 2011). Organizations are viewed as legitimate if their values and behaviors appear to be congruent with societal accepted norms (ibid). Organizations with high levels of legitimacy are argued to be able to survive longer, and can gain resources more easily (ibid). Supporters of the institutional theory argue that organizations need to adapt to external pressure since they need a societal mandate to keep their legitimacy. Simultaneously, they need to operate in manners that are conforming to societal expectations, and Boxenbaum and Jonsson (2010) assert that organizations often conform to rationalized myths that exist in society about what makes up a proper organization. Czarniawska and Guje (1996) argue that many organizations introduce the same type of changes around the same time that leads to organizations adopting similar practices, which is a phenomenon referred to as institutional isomorphism. Institutional theory holds the assumption that isomorphism leads to organizational legitimacy, which DiMaggio and Powell (1983) describe as an organization being acceptance by actor and other organizations in their external environment. Organizations' that conform to normative practices can be argued to do so because they want to show that they act in acceptable manners, by which external actors should view them as legitimate (Deephouse, 1996).

DiMaggio and Powell (1983) discuss three types of institutional mechanisms that influence how actors' legitimize the way they organize, for instance employer branding. *Coercive mechanisms* come from political influence and the issue of legitimacy. *Mimetic mechanisms* are standard responses to uncertainty. Mimetic mechanisms are imitations of strategies and practices that organizations implement since their competitors' use/have them (ibid). And the third mechanism is *normative* with mechanisms that are associated with professionalization, and explain that organizations' adopt certain practices because professionals within the organization claim they are superior. These mechanisms are different ways/attempts to reach legitimacy (DiMaggio& Powell, 1983). These three mechanisms are argued to have an impact on strategic HRM since they can cause HRM homogeneity across organizations (DiMaggio& Powell, 1983). In other words, looking at legitimacy can be done with the use of different types of framings that are described within frame theory, where organizations' motivate employees by framing the

organization in different ways (ibid). Framing is a way to justify and motivate actions, and can be seen as a part in the discussion of legitimacy as a type of legitimizing practice. Frame theory and institutional theory deal with legitimacy in a similar but somewhat different ways; the later is more theoretical whereas frame theory is more about the practice and can be used to get deeper into the active sense-making process that employees are involved in when legitimizing their actions and decisions.

#### 3.2 Frame theory

Frame theory presents a framework that can be used as a tool to organize peoples' experiences. For instance, Boundarouk et al., (2009) claim that frames help people to describe, explain, and predict events in their environment. Dexter, Perry and Berube (2007:np) describe frames as "a skeletal structure designed to give shape or support". Frames are not static characterizations, rather they change over time (Hallahan, 1999), and Boundarouk et al., (2009) claim that frames are context specific, flexible and have an interprevistic tone. Hallahan (2013) claims that frames can define problems, and also help define possible solutions when frames come into practice, a phenomenon referred to as framing. Framing is a type of practice, and is said to be a critical factor in constructing social reality as it helps people shape their perceptions, and provides them with context to be able to process information (Hallahan, 2007). The concept of framing (frames in practice) has received attention from various disciplines, and is used in many different ways (Hallahan, 2013). Framing builds on the notion that message producers are involved in the construction of social reality, and the meanings of a message are negotiated. A majority of psychologists argue that people use processes of association, and expectation, to make assumptions and assign meaning to a message, which may might not be evident in the message itself (ibid). Employer branding is about the associations that people relate with your company brand (Ambler& Barrow, 1996), and companies try to influence these associations by using different strategies.

#### 3.2.1 Different types of framing

Mosley (2014) discusses *brand framing* as being messages about the company as the employer of choice, and argues for the importance of keeping the content of the frame authentic and constantly updated in order to keep, and get, peoples' attention. Organizations are argued to have a tendency to use framing as a tool to analyze situations, and to justify actions that they take (Hallahan, 2013). *Organizational framing* has been used to study contemporary problems as it is

said that the "business contexts in which individuals work provide important cues that frame understanding of problems and lead to distinct behaviors" (Hallahan, 1999:211). Managers are said to be able to use **strategic framing** within the context of organizational communication as a strategy to enforce their version, regarding something, onto other members of the organization (Hallahan, 2007). In relation, Boundarouk et al., (2009: 473) argue that management literature shows that "it is the actor's perceptions of organizational processes, filtered through existing mental frames, which form the basis for the formulation and interpretation of organizational issues". Frames can be viewed as a tool to facilitate decision-making and problem solving (ibid). Further framing that can be referred back to strategic and organizational framing is the goal framing theory, as they all hold an organizational level of analysis, which is the level of analysis in this report as it aims to study the framing (practice) of employer branding within the public sector in Sweden.

#### 3.3 Goal Framing Theory

The goal framing theory embraces the concept of frames from the frame theory, and proposes that goals' frame the way people process information, and act upon it in specific situations (Lindenberg& Foss, 2011). The theory distinguishes between three overarching goals: the hedonic goal, which focus on immediate individual pleasure and is about achieving stimulation when performing a task. The *gain* goal that is about the desire to improve or preserve one's resources. The *normative* goal is about the desire to increase welfare for others (Lindenberg& Foss, 2011). The goal framing approach is built on the assumption that each one of the three types of goals' determines a certain structure of preference for the whole goal option (ibid). Multiple goals are argued to be active simultaneously, but the goal that is the strongest, most focal, is the goal that makes the frame, whereas the other goals' contribute to the frame but remain in the background. "In the foreground, the goal-frame structures the definition of the situation and works as a filter [...] to sort options" (Etienne, 2011:312). Actors are attentive and give priority to the front of the frame, whereas the background is not given the same amount of attentiveness. By understanding that people have these three different types of goals, an organization can work to manage so that the normative goal is in the forefront in employees' minds to get them focused on the goals of the organization (Lindenberg& Foss, 2011). Etienne (2011) claims that the goal framing approach is viewed as an interpretative tool while working with the sense making of empirical data.

The focus of this study will be on how a public sector organization practice employer branding, and in addition explore how they frame the alignment of employer branding with their overall goals as something legitimate. This study also aims to reveal how employees within a public sector organization describe meaning and make sense of employer branding.

## 4. Methodology:

#### 4. 1 Research strategy

This study takes on an explorative approach as it seeks to explore employer branding in the public sector, which is a relatively unexplored area with little previous research. A qualitative approach is suitable in order to get a holistic overview of peoples' feelings, interpretations and behaviors towards different situations (Hakim, 2000), which can help gain rich, descriptive, data of how employees' frame employer branding and interpret the meaning behind the strategy, Compared to quantitative research, qualitative research focuses on quality rather than quantity, depth rather than breadth, and is concerned with words rather than numbers (ibid).

#### 4.2 Research approach

The researcher of this report holds the perspective of employer branding as a concept that has been socially created, and to which actors assign meanings in different ways. Employer branding is not viewed as something that exists independent of social actors. Human actions are argued to arise from how peoples make sense of different situations (Easterby-Smith et al, 2012). Therefore it can be claimed that this thesis takes on an ontological position of constructionism (Bryman& Bell, 2007). Abductive reasoning is applied when analyzing the empirical data, which is a type of logical inference that allows the researcher to go back and forward in the data along the process of analyzing, which can give new insightful ideas. "Abductive reasoning about the data starts with the data and subsequently movies towards hypothesis formation" (Charmaz, 2009:103). Abductive reasoning may not always be correct, but it explore different types of hypothesis, which is why the method is argued to be able to come up with new and insightful ideas that come from a process of hypothesis-driven exploration (Kolko, 2010).

A single, explorative, case study is used as research approach. A case study can be used as a methodology to explore (Collis & Hussey, 2009), and seek to produce detailed and holistic knowledge about a certain phenomena (Eriksson& Kovalainen, 2008). Case studies take one or more selected examples from a social entity (Hakim, 2000), and entails a detailed exploration research approach of a specific case (Bryman& Bell, 2007). Employer branding will be explored within the public sector in Sweden by focusing on the specific case of the Municipality of Gothenburg to see how they practice and align employer branding with their overall goals and mission. Hakim (2000) claims that a case study can provide a descriptive detailed portrait of a social phenomenon. This approach is well suited for this report as it can help to provide detailed description of how one Municipality within the Swedish public sector work with employer branding in order to gain a deeper understanding of the phenomenon of employer branding within the public sector context.

#### 4.3 Case context

The Municipality of Gothenburg is built around ten different country district administration departments, together with more specialized departments and multiple state enterprises. In 2008, the Municipality of Gothenburg started a project, which they called "Attractive Employer", with the aim to develop strategies for employer branding (translated from Göteborgs stad, 2011). Different studies were performed with focus groups that were asked questions about their organization. The results from these studies were discussed, and three sentences were developed that describe their organization and what employees can expect to gain: 1) I work for citizens of Gothenburg to have good lives. 2) I am involved in the development of Gothenburg city. 3) I can choose between many different interesting jobs and career paths (translated from Göteborgs stad, 2011). The project included discussions about upcoming challenges, and three major challenges were identified; "we have satisfied employees, but not proud ambassadors". "The external image held of the organization as an employer". "A big group of employees will retire and leave a gap with vacancies that need to be filled with employees with the right competences" (ibid). The Municipality of Gothenburg came up with strategies for how to deal with these challenges, and their main employer branding strategy came to be about employees' knowing their mission, who they work for, and that they are expected to provide high quality services to all citizens of Gothenburg (translated from Göteborgs stad, 2011). The organization defined their main strategy into one sentence: we are here for the citizens of Gothenburg (ibid). The project

was finished up in 2011, and the outcome was different sub-strategies and a main strategy for employer branding that the organization developed in order to respond to the upcoming challenges in the near future.

#### 4.4 Research design

Research designs "are about organizing research activity" (Easterby-Smith et al, 2012). This report uses interviews as the design to organize employees' experiences, and understandings, of employer branding within the Swedish public sector context. Secondary data from databases, per reviewed articles, online journals, internal documents and books is collected in order to gain information about the research field (Collis& Hussey, 2009). Literature reviews is performed to gain in-depth understanding of the research topic in order to have an extensive understanding of what already have been studied within the field of employer branding. Primary data is generated from semi-structured interviews, which is a method concerned with exploring data of people understandings, opinions, and what they do, think or feel regarding different aspects (ibid). Semi-structured interviews enable the possibility to gain thick descriptions of employer branding. Surveys would not have given the same type of depth and understanding that this thesis aims to explore and gain. With the purpose of the study in mind, an interview guide was developed with some background question and then more specific about employer branding (see interview guide in the appendix).

#### 4.5 Data collection

#### 4.5.1 Sampling

This study focuses on employees working in different HR departments within the Municipality as the HR department is claim to be responsible for the implementation of employer branding within the Municipality of Gothenburg (Göteborgs stad, 2011). Purposive sampling and snowball sampling were used as the approach in the selection of whom to interview. Emails were sent out to HR managers, and the ones that declined were asked to send the interview request to some of their co-workers at their specific department. The Municipality of Gothenburg has ten different city district administrations, and the aim was to interview at least one individual from each city district to get a holistic view. However, the study ended up with including eight out of the ten city districts. Additionally, four people from four different specialized departments were interviewed. At three occurrences within the sample there were two from the same city district that were interviewed, but they held different roles within the HR department. Three respondents were men and the rest women, and the average age span among the respondents varied between early 30s to mid 50s.

#### 4.5.2 Interviews

Fifteen employees were interviewed, 11 face-to-face and four by telephone. Four interviews were conducted over the phone because the researcher and the respondents were geographically dispersed at the time that suited the respondents to be interviewed. The quality of the interviews performed over the phone held the same depth as the interviews that werer performed face to face. However, when comparing the content and depth in the answers, the researcher explored that the respondents that were interviewed over the phone tended to talk longer, and give longer answers to certain questions. The interviews were recorded after given consent from the participants. Additional notes were taken during the interviews, and after the interviews, thoughts and ideas that came up about the conducted interviews were written down in a journal right after each interview. The interviews were conducted and transcribed in Swedish, and took between 40-75 minutes, and they were transcribed after each interview had been conducted.

#### 4.6 Data Analysis

The empirical data is analyzed through thematic analysis, which is an approach that can help reveal underlying patterns existing in respondents' descriptions and answers during the interviews (Guest, MacQueen& Namey, 2012). Thematic analysis focuses on identifying and describing implicit as well as explicit ideas from within the empirical data (ibid).

The analysis was started by dividing the transcripts from the interviews into a document with two tables, in the first column all the questions were added, and in the parallel column was left empty. The interviews were read over several times, and key words (codes) from the transcripts were written down in the left column. These codes were written in Swedish and later on in the coding process translated into English. After the interviews had been read a couple of times, the left column was filled with key words and sentences that were sorted into categories with words and sentences that related to each other. These categories were named into different themes that were identified by code co-occurrence of the wordings in the left column. In this study themes are described as the underlying meaning in the answers from the respondents. In order to avoid deception the issue of creditability will be considered (Bryman& Bell, 2007). Creditability is

about ensuring that the researcher's understanding, and conclusion, regarding the research is correctly understood by getting confirmation of the members of the studied case. Deception is when the researcher represents the findings as something other than what it actually is (ibid). In order to keep the creditability of the data, the data is first coded into themes independently, which are presented in the result section. Secondly, the theoretical concepts are taken into consideration, which findings can be found in the discussion and analysis section. The theoretical concepts are left outside the first part of the analysis and coding process as the researcher holds the view that it is important to read the empirical data and see what it means in itself. Later on in the analysis, connections between the themes and the concepts from the chosen theories (institutional and frame theory) are discussed.

#### 4.6.1 Trustworthiness

There are many different arguments regarding the applicability of reliability and validity in qualitative research since the aims of qualitative research differ from the ones in quantitative data. Bryman and Bell (2007) discuss the approach of trustworthiness, which is an approach that can be applied for evaluating the value of qualitative data. This approach contains four parts that need to be taken into consideration during the evaluation; transferability, creditability, dependability, conformability (ibid). These four parts should be considered in order to create authentic, and trustworthy, data.

Instead of focusing on fulfilling the criteria of **transferability**, which is about the repeatability of the study (ibid), this qualitative study focuses on producing thick description of a social reality context. To keep the data authentic an interview guide is provided in the appendix in order to be transparent of the questions asked during the interviews. **Creditability** is about the consistency of the observation and the results, and in order to increase the creditability the interviews were recorded and transcribed in the exact words being said during the recording, the final paper is sent out to all participants in the study. **Dependability** evaluates how well the study achieves creditability and transferability (ibid). In other words, are the findings consistent and could they be repeated? **Conformability** regards the objectivity of the study to see to what extent the findings of the study are shaped from the respondents' answers or if the researcher has been biased and steered the findings in the way of his/her own interest (Bryman& Bell, 2007). In order

to avoid biases, limitations of the study have been considered, and description of how the analysis of data was performed is described in detail.

#### 4.7 Ethical considerations

Research ethics is crucial and ethical considerations such as informed consent, transparency to avoid deception, confidentiality, anonymity and protection of information are some of the research principals that will be applied in this report. After the sample of employees was identified, an email was sent out asking for participants willing to be part of this study (see appendix for letter of consent). The informative email included purpose of the study and considered ethical issues regarding the study. The participants were informed that their participation was voluntary, and that they could choose to avoid answering any of the questions, or stop the interview at any time if they wanted to. The email explains that the employees willing to participate in this study will be kept anonymous. Confidentiality issues were also included in the informative email, explaining that the participants' answers will be used for academic reasons only and that the data could not be traced back to the individual.

#### 4.8 Limitations of the research method

There are limitations that come with the choice to limit the study to one case, since multiple case studies are more robust and make it possible to compare and contrast the findings from the different cases (Eriksson& Kovalainen, 2008; Collis& Hussey, 2009). The aim of this research is not to compare, rather to investigate employer branding in a context with little previous exploration and contribute with descriptive empirical data. The findings from this report with the focus on one single case will make contributions of thick, descriptive data of employer branding within the public sector context. An additional limitation is the fact that the sample is rather small compared to the total amount of employees working with the organization. Additionally, the sampling method was not random but purposive sampling, which could have influenced the result of the study. Four interviews were conducted over the phone and Bryman and Bell (2007) discuss several advantages of telephone over personal interviews; they are cheaper, easier to supervise, and potential sources of bias is removed as the interviewer's personal characteristics cannot be seen as s/he is not physically present. However, there are also limitations that come with this type of interview as observation is out of the picture, which sometime can provide the researcher with rich data of how the respondent reacts, and feels, regarding certain questions.

When comparing the content and depth in the answers in the two different types of interviews, there was no significant difference. The transcripts from the interviews were translated from Swedish to English, and the translation can cause some levels of bias in the interpretation of some words/sentences since the meaning of some words might get lost in the translation process, which is a limitation of the study. However, the researcher went over the interviews several times to see if the words or sentences could be interpreted in another way.

# 5. Results

Findings from the empirical data show that consistency and sensitivity towards diversity are two important concepts for describing how the Municipality of Gothenburg practice employer branding. Two themes that evolved from analyzing the empirical data were; '*Organizing coherent employer branding*', which explains practices that aim for unification and consistency. And, '*Organizing employer branding sensitive towards diversity*', that deals with expectations of employer branding as a practice that is sensitive towards diversity. Employer branding seem to be about an act of balance between consistency and sensitivity towards diversity that employees describe as a daily dilemma, which they have to deal with.

#### 5.1 Organizing coherent employer branding

This theme concerns consistency and has three subthemes that aim for unification/coherence; *employer branding activities, match-making* and *the water drop effect*. The Municipality of Gothenburg is working hard to unify the image of the organization to bee seen as one unified entity. Employer branding is said to about associations that external actors have of the organization, and the Municipality of Gothenburg perceives it to be difficult to uphold the image of the organization as a legitimate actor when they lack consistency and coherence among its practices. Coherence is about having shared values and practices/processes that stick together, whereas consistency concerns the benefits to gain from behaving in the same way over time.

#### 5.1.1 Employer branding activities

Several of the respondents explain that they have difficulties to see the entire organization as one entity due to the wide range of operations, and the different opportunities that some respondents describe to exist between departments. The respondents explain that the Municipality of Gothenburg has made attempts to organize practices and processes within the organization so that they are performed similarly with consistency throughout the different departments. The main activity that is common for all departments (city districts, specialized departments and community enterprises) is advertisement, where the organization has a common framework for how to publish job offerings in newspapers, and other suited target-magazines. Respondents' explain that it was decided from the top down in the organization that everyone should have a common framework for their ads regarding job opportunities, as it was viewed to create separation when every department could choose how their ads should look like. Some departments have started to publish job openings on LinkedIn (a social media page), but the respondents' claim that the place of publication depends very much on the type of job offering, and where managers wish to publish it. Advertisement is an example of an employer branding activity where the organization's attempts to unify the image of the organization as one unified, legitimate employer. For instance as one respondent states;

"Because if you can see Municipality of Gothenburg as one big employer, the brand will become more permanent. Together we are stronger and there are economical benefits to gain, and it also has benefits for those that work here if it can become more transparent that we are one and do things in the same way" – HR specialist

Trainee program and internships were also mentioned as examples of employer branding activities that were conducted. Some departments within the organization are engaged in a trainee program where students get to experience working both in the public sector as well as the private sector for some set time period. The respondents' saw this as an important opportunity where they could provide the trainee with a good experience in order to strengthen their brand, as the trainee would be able to compare the organization to his/her experience in the private sector. Internship is another activity that can be viewed as an opportunity, where employees can provide students with an experience that makes them perceive the organization as legitimate. Respondents' from a few departments within the Municipality of Gothenburg state that they have started to provide opportunities for uneducated employees to get certifications and education which led to that employees can get a job title and a full time employment, instead of remaining employed as part time employees. As one respondent explains in the following quotation;

"They get their knowledge validated, which means that they can go less time in school, it looks a little different but if you have worked long and is considered to be proficient in certain areas you may not need to read those when you can have formal knowledge of these. These type of education form comes more and more in occupations where there is a lack of employees, you have to find other ways, not only those traditional school forms" – HR manager

One example of how the Municipality of Gothenburg market themselves to students is by being present at different work fairs held at different universities in Sweden, where they get to present themselves, and talk to students about the opportunities that exists within their organization. Respondents' perceive work fairs as a unique opportunity to communicate about aspects that makes them an attractive employer to prospective employees. However, few of the respondents' state that their departments are not present at work fairs because their produced material for how to present themselves is not ready to be communicated yet.

"We have many things that is common within the Municipality that we are using as we have said that we should work in these ways and developed similar practices to have a consistent approach and thereto related working materials" - HR manager

As the citation above shows, the Municipality of Gothenburg has developed common procedures for how to work within the organization should be conducted, and the HR function is organized around different processes that are similar for all departments. These procedures are suppose to function as support since it is decided that every department has a responsibility to adapt these practices to a level that best suits their specific department. The documents with common work practices and procedures/guidelines are described as vague, and many respondents' state that they want them to become more practical since they find it hard to know what to do in practice. These documents had been developed with the aim to make sure that everything is standardized and done correctly, and to make sure that all departments work in alignment with the overall mission and goals of the organization. However, the majority of respondents' perceives that these procedures are very abstract. For example as one respondent states; "We are good at producing documents and material to use but these things needs to be taken into practice and not stay in the written documentations" - HR specialist

#### 5.1.2 Match-Making

During the interviews, respondents' discuss the importance of changing their way of thinking, and argue that many put too high requirements in their advertisement text. Some respondents' suggest that the recruitment process should work as some type of match-making where the applicant describe his/her attributes and experiences, and what s/he likes to work with and why, and then the recruiter could offer some different types of jobs that fit with their description. This is something that respondents' argue could be beneficial for current employees as well, that could be matched to a different job/department within the Municipality when employees' feel that they have fulfilled their current job position and are in a need of change. This match-making was discussed as a suggestion to provide alternative careers within the organization instead of having employees quit and leave the organization. Following quotation indicate that values are important aspects to consider when trying retaining employees as well in the attempts to attract;

# *"It is about values. And the majority of employees outside our organization are value-driven. And as a public organization we have moral values I would say" - HR manager*

Many of the respondents said that they believe it is more important to focus on the social attributes of the applicants rather than their experiences and educational background. Most employees held the perspective that it is important that applicants are motivated and have social values that fit with the values of the organization. This was argued to be an important aspect to consider, as the applicants potentially were about to join an organization that is humanistic driven with a mission to achieve the common good for all citizens. The mission of the organization is something that the majority of respondents' claim that they carry with them, something they have in the back of their heads and reflects back to every now and then. Respondents view the work within the municipality as tough and difficult, especially the transparency of the expectations that are placed on the individual employee. Some respondents argue that applicants should be informed about the expectations before they decide to accept a job offer. This in order to listen to if the applicant is ready to take on that type of responsibility, so that there is some type of match during the contact of the recruitment process. If the applicant

is not informed about the expectations during the recruitment, the applicant could possibly quit when his/her expectations of the job were not fulfilled, which few employees stated would be a wasted recruitment. Some respondents explain that if applicants get a negative experience when they are involved in a recruitment process, the individual can spread negative words about their experience, which can cause issues for the organization.

#### 5.1.3 The water drop effect

From respondents' descriptions, the individual employee seems to get a lot of responsibility, as s/he needs to make sure that his/her performance is strengthening external actors' associations of the organization as the employer of choice. It is said that every employee is expected to contribute and help spread positive aspects and associations to help strengthen the positive image of the organization. The responsibility on each individual employee to make contributions can be seen to be about unity as the municipality strives to make sure that employees' are operating towards the same goals and work to create a positive reputation to make the organization viewed as legitimate by external actors. The following citation indicates that the respondent believes it is important to consider the employer branding work to be part of their every day work;

"Its practical, you have to perform it. I mean that the attractiveness comes in the offering as a whole, in the things that you do every day" - HR manager

Respondents claim that satisfied employees will talk good about their work at among family and at dinners with friends, which they believe can help spread positive associations of the organization to external actors. Respondents' working within HR argue that it is important to work hard to create a good working environment, based on their assumption that a good working environment creates satisfied employees, which one of the respondents' statement indicates;

"I use to say that if we concentrate, and can praise, our thousands employees at our department, and makes sure to take care and develop our current employees, that we are clear about expectations and transparent what they can expect from us, that we work to make them feel pride, then we have done a big contribution" - HR manager

#### 5.2 Organizing employer branding sensitive to diversity

Another prevalent theme that evolved when examining the empirical data was about employees dealing with vagueness and ambiguity. This theme has three subthemes; *humanistic perspective, frustration over the abstract and visional* and *ambiguity*. The humanistic perspective of the company, and the organization's mission to be accountable for the common good for all, are some things that the respondents' experience as vague, which makes them unsure of how to perform in different situations. The respondents agree that they have meaningful professions but that the overall mission is vague, which makes it difficult for them to understand what they should do in order to remain legitimate. The respondents' describe their responsibility, and their work of trying to balance diversity with consistency as a prominent challenge, which they have to battle as it is part of their everyday work life.

#### 5.2.1 Humanistic perspective

"The feeling of helping others, and that someone else appreciate what you do for them is amazing... It is good to feel that you are there for someone and that the things you do make a difference" - HR specialist

The respondents mention the humanistic perspective of the organization to be the key that makes their business unique and attractive. At the same time, some respondents describe it as a challenge that they need to get better at communicating and informing applicants about so that they are prepared that they might face situations where they hold feelings of performing insufficient work. Some employees in this study explain that it is challenging to never feel that they get to accomplish and finish something, which some respondents perceive cause feelings of frustration. One respondent states;

"When you work with products, you might put together two pieces and then the work is done. When you work with humans you can never really finish the work" - HR specialist

The extent to which the different departments' practice employer branding varies, and some respondents state that they look at what other departments do to get inspired and ideas for what

to do. Respondents' deal with the uncertainty by imitating what others does since they have to have something to relate to when taking decisions. The empirical findings suggest that employees' seem to make sense of employer branding by making comparisons and justifications, and also by imitating employer branding activities conducted by other departments, or the private sector.

"It is harder and more comprising to work with employer branding in the world of public sector organizations. When you talk about employer branding you associate it very much with the work around a brand. Here the brand is Gothenburg city and a lot of different departments, then it is little harder to identify compared to when you have one brand that is famous for a product or service" - Communications manager.

The quotation above shows that employer branding can be perceived as a difficult practice. However, employer branding is regarded an obligation, which needs to be embraced in order to remain competitive on the labor market. Employer branding is often described by the respondents in comparison to the private sector combined with vindications of what makes them unique. Some of the respondents claim that they perceive that they are behind the private sector regarding how they practice employer branding. Some employees even make the claim that the private sector is better at employer branding. However, after making such claims, respondents defend themselves by arguing that they have more meaningful work, and that the public sector wins over other businesses in the long run, as some of the respondents claim that many ends up working in the public sector.

#### 5.2.2 Frustration over the abstract and visional

Even tough most respondents agree upon the fact that it is the human perspective that makes their business different to others, many employees find it difficult to know what to communicate to appeal external actors. The majority of respondents stress the importance of living up to the communicated messages in order to remain legitimate. As the following citation indicates;

"You cannot just live on the fact and state 'hey come to us, because you can do good here', you actually have to show what you do, so it's practical, it has to be performed" - Communications manager

Respondents' explain that it is important to be able to live up to what is being communicated, but this is described to be a challenge since employees work with abstract guidelines, which that makes it hard for them to know how to plan their practices and upcoming activities to make sure that they are aligned with the mission and overall goals of the organization. Much responsibility is being put on the individual employee that needs to take decisions, which are simultaneously consistent with the organization's guidelines, and sensitive towards societal diversity. The challenge seems to be to keep the practice of employer branding consistent with continuous activities, and some of the given explanations for this were lack of time, having few opportunities and other tasks that were more important to focus on. Some respondents even argued that the available opportunities in the departments differed, which makes it hard for everyone to have the same opportunities to work with employer branding.

#### 5.2.3 Ambiguity

"It is more about the goals and not any plans, no summaries and activities that can be implemented. It is more objective than practical. It is up to every department to implement activities, and it would have been fun to get inspiration from others since it is difficult to push this issue yourself" - HR specialist

This quotation shows that even though the respondents are aware of that they are suppose to adapt the common guidelines to a level that suits their department, some respondents perceive the balance between following the common guidelines, and the adaption to make it fit their specific department, to be a challenge that employees' battle with. Regarding the decision about having a unified framework that everyone should follow when developing ads, is a decision viewed as positive by some of the respondents, whereas few respondents argue that the similarity made the organization loose the value of diversity that exists among the different jobs and within departments. One respondent states;

"I would almost say that there are ten culture, and we are very much characterized by the areas where we operate, and I do not believe that the ambition to create a common culture would, it is *neither possible or would be good I think, since it looks so different therefore you have to adapt a little to the difference"* - Communications manager.

The vision of the organization is political and respondents' describe it to be too visional as they claim that they had a hard time to get a grip of. The surrounding environments in where the departments are located look very different from each other, and therefore the respondents believe that it is important that the work practices at each department are adapted so that it suits the citizens living within that specific area of Gothenburg. The challenge to be both unified and diversified is important and can be regarded to be about legitimacy. Every citizen has to be treated equally and get the same amount of opportunities, and since the mission of the Municipality of Gothenburg is to serve common good for all citizens, the respondents' claim that it becomes extremely important for them to be sensitive towards diversity in order to avoid to be regarded as unfair. The following quotation indicates that sensitivity towards diversity is an important aspect that is combined with the idea that everyone should be treated equally;

"We are trying to implement a more systematic ways to work when it comes to the recruitment process in the case that we want managers to use interview guides in order to secure that everyone that applies gets an equal treatment, and have the same opportunity to get through at the interview occasion" – HR specialist

The empirical findings show that respondents' hold the view that people are different from each other, and when employees are expected to treat everyone equally in order to be regarded as being fair, it becomes problematic to simultaneously be sensitive towards diversity. During the challenging work to standardize the recruitment process within the organization, fairness is considered as an important part, as the organization strives to perform the process equally across the different departments in order to be seen as legitimate actors.

# 6. Discussion and analysis

Employer branding is a concept that has gained more attention as organizations acknowledge the importance to improve their capabilities to attract and retain qualified human capital. Most research of employer branding is from the private sector, and researchers almost seem to take for granted that employer branding can be performed similarly in all work spheres, but the results of this study shows that this may not be the case.

The purpose of this study is to explore how the public sector in Sweden practice employer branding in order to make the alignment between employer branding and the organization's overall goals and mission legitimate. This study focus on the case of the Municipality of Gothenburg within Sweden, and the results from the empirical findings show that the municipality frame employer branding as a strategy that helps them deal with issues of legitimacy. Consistency and sensitivity towards diversity seem to be two important concepts that explain how the municipality chooses to practice employer branding. The act of balance between these two ways of organizing (organizing coherent employer branding, organizing employer branding sensitive towards diversity) can be explained by how the Municipality of Gothenburg has decided to align employer branding with the overall mission and goals of the organization.

#### 6.1 Organizing coherent employer branding

The first research question that was answered in the result section was about how public sector organizations practice employer branding, which was answered by looking at how employees within the municipality frame employer branding. The Municipality of Gothenburg practice employer branding through having a unified framework for their ads, and a structured recruitment process that aims for consistency. The organization strives for consistency by developing certain practices within the organization coherent in order to ensure to be seen as one, legitimate, actor on the labor market. The respondents' perceived the situation of having each department within the organization decided by themselves how their ads should look like as a problem, therefore standardized procedures were developed for the whole organization, which employees are expected to adapt to in order to fit to their specific department. Organizational framing, which Hallahan (1999) describes as a strategy to use to understand issues and problems as the business context can provide cues that lead to distinct behaviors, could be argued to been

used as a strategy in the decision about making the recruitment and advertisement more structured in order to unify the organization. Strategic framing can also explain the situation of the organization deciding upon the common procedures/guidelines. The organization communicates about the importance to practice employer branding, and provide guidelines and expects that employees take responsibility to implement employer branding activities at their specific department. This could also be described as some degree of institutional coercive mechanism that the municipality adapts in order to legally secure that every department are following policies/guidelines. DiMaggio and Powell (1983) describe coercive mechanisms to have a political influence, and can be pressures from other organizations or governmental mandate (ibid). However, the broad span of opportunities within the frame is something that the respondents experience as vague and too ambiguous, and describe to be too abstract for them to know what to do in terms of activities that are related to employer branding. The majority of respondents' believe that they have vague guidelines for how to develop employer branding activities that aim to strengthen the image of the organization as the employer of choice.

Employees explain feelings of being uncertain of how to practice employer branding, and the majority of respondents from different departments said that they often were looking at what other departments within the organization do to get ideas and inspiration from their employer branding activities. Employees use some type of mimetic mechanisms that is described within institutional theory (DiMaggio& Powell, 1983), as the majority of employees seem to take ideas of activities from other departments within the Municipality. This type of framing where employees make sense of employer branding as an activity that they can perform by imitating other departments reminds about the mimetic institutional mechanism that DiMaggio and Powell (1983) describe, and discuss in relation to issues of legitimacy. Employees respond to the guidelines, that they experience as vague, by taking decisions to perform activities that other departments do, as the uncertainty takes over and makes it difficult for them to take decisions by themselves. Respondents described that they also were looking at what the private sector was doing in terms of employer branding. In this study these behaviors and descriptions of when employees look at what other departments perform, and the private sector, will be referred to as mimetic framing. Hence, employees' use mimetic framing as a tool to make sense of employer branding, something they can use in order to deal with their uncertainty. In other words,

respondents deal with their described uncertainty by following similar paths that other department takes regarding how they choose to practice employer branding. The respondents behavior can be regarded as the institutional theory's described concept of isomorphism (Deephouse, 1996; DiMaggio& Powell, 1983), as employees seem to assume that the activates that other departments and sectors conduct are legitimate, since they decide to perform similar practices themselves.

Further, respondents' explain that they believe it to be crucial for them to embrace employer branding as the labor market is showing a lack of talents and recruitment challenges. The respondents frame employer branding as a concept that they have to embrace, and take decisions about, no matter how difficult it might be. This in order to avoid fall behind other businesses that practice employer branding as a strategy for how to better attract talents. Respondents from different departments within the Municipality of Gothenburg state that they have started to provide job opportunities for employees that do not have education to provide them with the opportunity to get certificated and the chance to get a job title. This could be viewed as an activity that the organization execute in order to be regarded as legitimate employer. Dyhre and Parment (2013) claim that companies need to start communicate about the different available career paths available, since research that has shown that an employer get more attractive in the eyes of prospective employees if the career paths within the company are easily identifiable. Opportunities to make career or change job within the organization was something the respondents' highly valued. Respondents' believe that the organization should work to get better at communicating the existing opportunities for these to become more transparent for external actors, as well for current employees. Respondents' statements go along with Backhaus and Tikoo (2004:511) argument that without opportunities for advancement the organization loose its competitive edge and become less attractive for applicants.

#### 6.2 Organizing employer branding sensitive towards diversity

The theme "frustration over the abstract and visional" is similar to the description of brand framing, where Mosley (2014) argues for the importance of keeping the content of the frame authentic and constantly updated in order to keep and get peoples' attention. Mosley (2014) describes brand framing as a strategy for how to frame messages about the company as the employer of choice. Employees' believe it is important to perform activities that make sure that

what is being communicated actually is conducted in order to ensure to keep their legitimacy. This goes along with the claim that Dahlqvist and Melin (2010) make of the importance for public sector organizations to be able to show external actors that they deliver services that contribute to the common good for all citizens in order to gain acceptance, and keep their legitimacy. If the Municipality of Gothenburg wants to communicate messages about being the employer of choice, employees feel that they want to be able to take these messages into practice in order to uphold the legitimacy of the messages that are being communicated. These statements are similar to the claim that Christiaans (2013) makes about the communicated benefit that should paint up an accurate image in order to avoid developing unrealistic expectation among prospective employees, since unmet expectations in the recruitment process might lead to problems in the future (ibid).

Respondents emphasize the importance of informing employees regarding different matters in order for them to know what is going on so that they can take pride in the work they do. The assumption is that if they know what goes on, employees will get more satisfied and a positive attitude towards work will start to build. Employees that hold a positive attitude towards work will help spread positive associations about their workplace, which can contribute to the positive reputation of the organization, like a water drop effect; spreading from the inside out. Van Mossevelde (2012) claims that companies should take on an inside-out approach with the assumption that happy employees will function as proactive brand ambassadors, and contribute to a positive reputation of the company. This is something that is discussed in theme "the water drop effect", where employees are explained to have a responsibility to make contributions, and function as ambassadors for their organization. Dahlqvist and Melin (2010) argue that many organizations within the public sector have reputation issues, which was something many employees with the Municipality of Gothenburg mentioned, as they find themselves working in a constant uphill, and some respondents' blame media for having some part in this as they claim that media tends to write about the negative aspects, but not so much about all the good things that they do within the organization.

Edwards (2005) claims that an individual who identifies with the organization is very likely to accept the goals and values of the organization, which usually results in that they choose to

remain within the organization. Employees within the Municipality agree that it is the human perspective that makes their organization attractive and separates them from other businesses. The normative goal was well established, which was expected, as the mission of the organization is to serve the common good for all citizens of Gothenburg. The normative goal within the goal framing theory is about the desire to increase welfare for others (Lindenberg& Foss, 2011), which is what the description of the mission of public sector organizations. This can be seen through the Municipality's way of organizing employer branding as a practice that pays attention and is sensitive towards diversity. The gain goal is about the desire to improve or preserve one's resources Lindenberg& Foss (2011), which was another type of goal that employees used to frame their understanding of employer branding. Employees' claim that it is important to be able to compete for the best qualified talents in order to be able to attract them to their organization, which can be described as a desire to preserve ones resources to be able to continue deliver high quality services to citizens. The respondents' claim that they have the opportunities to be able to compete, but that the issue is that they need to get better at communicating about these to external actors, as well to current employees. Another type of goal that was revealed within the findings came from statements and comparisons with the private sector. The theme "humanistic perspective" shows that employees view employer branding as an important concept to embrace, it also show how the respondents' frame employer branding as a concept that the private sector manage and work with much better. Some of the respondents claim that private sector organizations are much better at employer branding, and few argued that they believe they have to start work more like the private sector. Continuous comparisons with the private sector show characteristics of another type of goal, a goal that can be characterized as having mimetic nuances, which could be described as some type of mimetic goal. This comes back to the mimetic framing that departments are involved in when they imitate employer branding activities that the public sector and different departments conduct.

# 6.3 Organizing coherent employer branding sensitive towards diversity - A dilemma?

The second research question of this report, which is about the alignment between employer branding and the overall goals of the organization, is answered by the empirical findings that shows that the biggest challenge respondents describe is regarding employer branding as an act of balance between consistency and diversity. The expectations of having to be accountable and manage the common good for everyone, and at the same time balance practices so that they are sensitive towards diversity and aim for consistency is something that is hard to accomplish. This act of balance is specific to each unique situation that employees face, which explains why the guidelines and vision provided from the organization are viewed as abstract by the employees, since it is impossible for the organization to be specific on what to do since each situation is unique in its own way. This is something that the respondents described to be very difficult as they are left with responsibility to decide on activities to perform with no direct procedures for how to act when they face situations that require decision-making in the manner of having to act and balance the decision between diversity and consistency.

# Organizing coherent employer branding

- Employer branding activities
- Match-making
- The water drop effect

## Organizing employer branding sensitive towards diversity

- Humanistic perspective
- Ambiguity
- Frustration over the abstract and visional

Model modified by the author of this thesis.

G

0

Α

L

In the public sector where the mission is to serve the common good of all citizens, it is important to respect diversity and treat everyone equally, which explains why Municipality have aligned employer branding with their assignment in a fashion that is more visional and less direct. People are different from each other, and when you are expected to treat everyone equally and have unified practices that are consistent, in order to be regarded as being fair, it becomes problematic to simultaneously be sensitive towards diversity. These two ways of organizing employer branding are not compatible and are causing tensions to exist, which makes it hard to locate employer branding activities in the middle. It is a challenge that creates a dilemma for employees that has to perform work tasks that are sensitive to an external diversified world and functions in consistent unifying manners. These two types of organizing activities and decision-makings have underlying tensions that are challenging to deal with, and cause uncertainty among employees.

The challenge becomes to be about legitimacy as the actors always have to be on their guard when taking decisions and avoid creating attention from for example media that might want to publish a story about something that turned out to be more a more difficult decision-making then what was expected. Employees within the Municipality of Gothenburg are experiencing issues of legitimacy as they are trapped in a dilemma that involves issues of balancing decisions with two themes or organizing that are not really compatible with each other. These challenges influence the way employees' make sense of employer branding. The act of balance between consistency and diversity describes how the alignment between employer branding and the overall mission looks like within the Municipality of Gothenburg. Coherence and consistency are used as a strategy to align the employer branding practice and goals with the organization's overall mission to make sure that the organization practice employer branding in manners that pays attention and is sensitive towards societal diversity. The goal is to conduct activities, and practice employer branding that are balance in the middle, which is where the Municipality aims to be regarding how they practice employer branding.

## 7. Conclusion

Exploring how the Municipality of Gothenburg practice employer branding reveals issues of legitimacy. The findings suggest that employees within the Municipality of Gothenburg want to be viewed as legitimate, which the employees try to achieve by using mimetic framing and the strive to fulfill mimetic goals. The municipality chooses to solve the issue of legitimacy, involved in the act of balance, by using strategic framing. Whereas employees' attempt to solve the issues by using mimetic framing, which could can be argued to not really be regarded as a strategy as they decide to do what others do. Employees are responsible for performing legitimate practices, which is hard when the guidelines/common procedures are vague and abstract. Most respondents solve the battle by imitating what other departments do since they do not know what to do, instead they rely that since other have done these activities, they must be viewed as legitimate. The employees need to have something to relate to, and the risk from this

imitating behavior might cause the alignment and the act of balance between unity and diversity may get lost in the process of imitating others. The balance between consistency/unity and diversity is something that employees within the public sector are dealing with, therefore an existing dilemma can be argued to exists, that is characterized with no direct solutions since every day is different from each other. The Municipality of Gothenburg strives to have internal coherence and external sensitivity towards diversity. The dilemma arises when the organization is supposed to have an internal coherence and at the same time be sensitive towards a diverse society. The dilemma can explain why the vision and guidelines are experienced as abstract by the employees, hence, the organization cannot be too detailed and structured in the provide guidelines/procedures in order to strive to secure the balance act between consistency and diversity. In contrast, if the organization unifies processes and practices and makes them too coherent, the value of diversity may be lost. The question of how to manage and organize a public sector organization is involved in the ongoing discussion that is related to new public management.

Citizens expect that they will be provided with public services, therefore public sector organizations have to take decisions no matter how difficult it is. Therefore the results from this case study of one Municipality in Sweden can be assumed to be found in other municipalities within the public sector. Public sector organizations are not like private sector organization and the work for employees within the public sector becomes very difficult when the work is organized in a way that creates confusion and leaves employees' with obligations to deal with uncertainties. The findings suggest that the underlying factor in the discussion is about what specific opportunities the individual employee possesses, because when employees are aware of the available opportunities, they know what to expect.

This report contributes with its empirical evidence about the act of balance that creates a dilemma for employees within the public sector in Sweden. Complementary, the report makes a small theoretical contribution as a new type of framing was discovered (mimetic framing and mimetic goal), which could help develop, and expand, the field of frames and frame theory. The empirical findings of the dilemma create awareness that this is a problem that needs to become more transparent.

# 8. Final concluding remarks

The issue of balancing diversity with consistency is a dilemma that will always exists, therefore it is important to acknowledge this dilemma that employees' within the public sector experience. Additionally, employees should be aware about the dilemma and about the existing job opportunities, and that these types of situations where they have to balance between sensitivity towards diversity and consistency are responsibilities that they will have to accept. This is were HR can take on a more supportive role and function as leaders by helping employees manage this reoccurring dilemma. A recommendation for the HR function is to communicate about these results so that management becomes aware about the uncertain situation that employees battle with. In addition, there should be supportive opportunities provided from HR that needs to take on a leadership role and make sure that the organization educates their employees about this dilemma and what is expected of them. A support function such as HR becomes extremely important in these types situations where employees feel uncertain that they cannot deliver what is expected from them.

## 8.1 Suggestions for future research

Further research could look into what ways HR can function as a support system for employees within the public sector, in their daily battle with the dilemma of balancing the practice of employer branding. That type of study would be of interest as HR could proactively work with employees to better deal with the dilemma, which might become less prominent.

Another suggestion would be for future research to add to the empirical findings of this research to expand the research field of employer branding within the public sector, which would be of important in order to enhance further understanding. This study was conducted within one municipality in Sweden, therefore further research could be conducted by explore other types of public sector organizations, as it would be of interest to explore if the dilemma found within in this study can be found elsewhere. In addition, research could explore what specific sectors within the municipality in which employees' battle with this dilemma, as the dilemma might be more prominent in some specific sectors. This study found a new concept that is referred to as mimetic framing, which was used by respondents to make sense of employer branding, therefore an additional suggestion would be investigate the concept to explore if it exists in other contexts.

## 9. References

Aggerholm, H.K., & Thomsen, S.E.A.C., (2011). Conceptualising employer branding in sustainable organizations. *Corporate Communications: An international Journal* Vol. 16(2), pp.105-123

Ambler, T., & Barrow, S. (1996). The Employer Brand. *Journal of Brand Management* Vol. 4(3), pp.185–206.

Backhaus, K., Tikoo, S. (2004). Conceptualizing and researching employer branding. *Career Development International* Vol. 9(5), pp.501-517.

Barrow, S., & Mosley, R. (2005). *The Employer Brand- Bringing the Best of Brand Management to People at Work*. Chichester, England: John Wiley & Sons Ltd.

Blumberg, B., Cooper, D.R., & Schindler, P.S. (2011). *Business Research Methods*, (3<sup>rd</sup> ed.) Berkshire, UK: McGraw-Hill Education.

Boxall, P. (2008). "Strategic Human Resource Management" in Clegg, S.R., & Balley, J.R (Eds), *International Encyclopedia of Organization Studies*. SAGE publications, Inc.

Boxenbaum, E., & Jonsson, S. (2010). "Isomorphism, Diffusion and Decoupling", in Greenwood., R, Oliver, C., Suddaby. R., Sahlin, K (Eds), The SAGE Handbook of Organizational Institutionalism. SAGE Publications Ltd. pp. 79-99.

Bondarouk, T., Looise, J.K., & Lempsink, B. (2009). Framing the implementation of HRM innovation: HR professionals vs line managers in construction company. *Personnel Review* Vol. 38(5), pp.472-491.

Bryman, A., Bell, E. (2007). *Business research methods,* (2<sup>nd</sup> ed). New York, United States: Oxford University Press Inc.

Bryman, A. (2012). *Social research methods* (4<sup>th</sup> ed). New York, United States: Oxford University Press Inc.

Charmaz, K. (2009). Constructing Grounded Theory: A Practical Guide through Qualitative Analysis (Introducing Qualitative Methods series). London: SAGE Publications Ltd.

Chhabra, L. N., & Sharma, S. (2014). Employer branding: strategy for improving employer attractiveness. *International Journal of Organizational Analysis* Vol. 22 (1), pp.48-60.

Christiaans, L. (2013). International Employer Brand Management: A Multilevel Analysis and Segmentation of Students' Preferences. Weisbaden: Springer Gabler. pp.1-264.

Collis, J., & Hussey, R. (2009). Business Research – A practical guide for undergraduate & postgraduate students (3<sup>rd</sup> ed). UK: Palgrave Macmillan

Czarniawska, B., & Sevon, G. (1996). *Translating Organizational Change*. Berlin: de Gruyter. pp. 1-284

Dahlqvist, U., & Melin, F. (2010) Varumärken i offentlig tjänst. Malmö: Liber AB

Deephouse, D.L. (1996). Does Isomorphism Legitimate? *The Academy of Management Journal* Vol 39 (4), pp.1024-1039.

Dexter, R., Perry. S & Berube, W. (2007). "Frame theory", in English, F.W (Ed), Encyclopedia of Educational Leadership and Administration. Retrieved February 25, 2015.

Dyhre, A., & Parment, A. (2013). *Employer Branding: Allt du behöver veta för att bli en attraktiv arbetsgivare*. Malmö: Liber AB. pp.3-180.

Edwards, M.R (2005). "Employer and Employee Branding: HR or PR?", in Bach, S. (Ed), *Human Resource Management: Personnel Management in Transition*, Blackwell, Oxford. pp. 266-286

Edwards, M.R. (2009). An integrative review of employer branding and OB theory. *Personnel Review* Vol. 39(1), pp.5-23

Edwards, M.R., & Kelan, E.K (2011). "Employer Branding and Diversity: Foes of Friends?" In Brannan, M.J., Parsons, E., & Priola, V. (Eds), *Branded Lives – The Production and Consumption of Meaning at Work*. pp.168-181. Cheltenham: Edward Elgar

Eriksson, P., & Kovalainen, A. (2008). *Qualitative Methods in Business Research*. London: SAGE Publications Ltd. London: SAGE Publications Ltd

Easterby-Smith, M., Thorpe, R., Jackson, P. (2012). Management Research, 4<sup>th</sup> ed.

Etienne, J. (2013). Compliance theory: A Goal Framing Approach. *LAW&POLICY* Vol. 33(3), pp.205-333.

Fisher, I. (1997). Locating Frames in the Discursive Universe. Sociological Research Online, Vol. 2(3). Retrieved February 24, 2015 from: <u>http://socresonline.org.uk/2/3/4.html</u>

Foster, C., Punjaisri, K., Cheng, R. (2010). Exploring the relationship between corporate, internal and employer branding. *Journal of Product & Brand Management* Vol. 19(6), pp.401-409.

Guest, G., MacQueen, K.M., Namey, E.E.(2012). Applied Thematic Analysis. Thousand Oaks, United States: SAGE Publications, Inc.

Göteborgs stad. (2011, january). *Slutrapport Employer Branding*. Unpublished intranet document.

Göteborgs stad:A. (nd). Kommunens organisation. Retrieved march 15, 2015 from: http://goteborg.se/wps/portal/invanare/kommun-o-politik/kommunensorganisation/!ut/p/b1/jYtLCsIwFADP4gHse\_k0eVmmQv\_0IwptNlJFSqFpN6LXtx5AdHYDM-Cg3zMmhSYISEIHbhme0zg8pnUZ5o87dWIY3ILELNaJiTE7FU1cFWUdNXwLi04JDaVukSkMgkxs-n5aFoh0Ir\_fvyCxV9\_Dm66uB18wEGnIeaOBIDWpJRRkOVrv4O3s0xjXb3BjeXhII!/dl4/d5/L2dBISEvZ0FBIS9nQSEh/

Göteborgs stad:B. (nd). Förvaltningar. Retrieved march 15, 2015 from: http://goteborg.se/wps/portal/invanare/kommun-o-politik/kommunensorganisation/forvaltningar/!ut/p/b1/jYtLCsIwFADP4gHse2nS5mUZC236oT8UbDZSRUqgaTei1 7ceQHR2AzNgYdgzJrikmJOAM9hlfLppfLh1GeeP2\_jSsqKjA9PYZCrF\_Fi2aV1WTZ\_wLRi2I Mm0EbJCpCqLMNfm1KuOc9T8vx-\_oPHXX4B1Vx-8bj7AIAwjSSEpRVKQipWE2qz-Dt7OKU169wZUo8RZ/dl4/d5/L2dBISEvZ0FBIS9nQSEh/#htoc-0

Hallahan, K. (1999). Seven models of framing: Implications for public relations. *Journal of Public Relations Research* Vol. 11 (3), pp.205-242.

Hallahan, K. (2007). *Strategic Framing*. Retrieved February 24, 2015 from: <u>http://www.pitt.edu/~mitnick/MESM10/HallahanStrategicFramingHallahan\_ICE\_051507.pdf</u>

Hallahan, K. (2013). *Framing Theory*, in Heat, R.L (Ed), Encyclopedia of Public Relations. Retrieved February 24, 2015. DOI: <u>http://dx.doi.org.ezproxy.ub.gu.se/10.4135/9781452276236.n202</u> pp.361-364.

Hylen, E. (2011). *Kommuner och Landsting kämpar mot privata sektorn och gamla fördomar*. Retrieved January 24, 2015 from: <u>http://www.dagenssamhalle.se/debatt/kommuner-och-landsting-kaempar-mot-privata-sektorn-och-gamla-foerdomar-257</u> Johnson, G., & Scholes, K. (2001). *Exploring Public Sector Strategy*. Edinburgh Gate Harlow, England: Pearson Education Ltd, pp. 1-331.

Kolko, J. (2010). Exposing the Magic of Design: A Practitioner's Guide to the Methods and Theory of Synthesis. Oxford University Press. doi:10.1093/acprof:oso/9780199744336.003.0004

Lalwani, N., & Parmar, A. (2012). An empirical study on external and internal employer branding. *International Journal of Research in Social Sciences* Vol. 2(3), pp.167-179.

Lindenberg, S., & Foss, N.J (2011). Managing joint production motivation: the role of goal framing and governance mechanisms. *Academy of Management Review* Vol. 36 (3). pp.500-525

Lindmark, A., & Önnevik (2011). *Human Resource Management; Organisationens hjärta*. Lund: Studentlitteratur AB. pp.7-351.

Magnusson, P. H. (2011) "Pensionavgångar - nådastöt eller nystart?", in Komlitt (ed), *Talanger sätter ribban – vinnande personalstrategi för offentliga sektor*. Höganäs: Komlitt AB, pp.15-20.

Martin, G. (2009). Driving corporate reputations form the inside: A strategic role and strategic dilemmas for HR? *Asia Pacific Journal of Human Resources* Vol. 47(2), pp.219-235.

Martin, G., Beaumont, P., Doig, R., Pate, J. (2004). Branding: A New Performance Discourse for HR? *European Management Journal* Vol. 23(1), pp.76-88.

Martin, G., Gollan, P.J., Grigg, K. (2011). Is there a bigger and better future for employer branding? Facing up to innovation, corporate reputation and wicked problems in SHRM. *The International Journal of Human Resource Management* Vol. 22(17), pp.3618-3637

Minchington, B. (2010). *Employer Brand Leadership – A Global Perspective*. Torrensville: Collective Learning Australia.

Moroko, L., & Uncles, M.D. (2008). Characteristics of successful employer brands. *Journal of Brand Management* Vol. 16(3), pp.160-175

Mosley, R. (2014). "Employer Brand Management", in Kompella (Eds), *The Definitive Book of Branding*. New Delhi, India: Sage Publications Inc. pp.217-240.

Najeeb, A. (2014). Institutional theory and human resource management, in H. Hasan (Eds), *Being Practical with Theory: A Window into Business Research*. Wollongong, Australia: THEORI, pp.25-30.

Näppä, A., Farshid, M., Foster, T. (2014). Employer branding: Attracting and retaining talent in financial services. *Journal of Financial Services Marketing* Vol. 19(2), pp.132-145

Paauwe, J., & Boselie, P. (2003). Challenging 'strategic HRM' and the relevance of the institutional setting. *Human Resource Management Journal* Vol. 13(3), pp.56-70.

DiMaggio P.J. & Powell W.W. (1983). The Iron Cage Revisited: Institutional Isomorphism and Collective Rationality in Organizational Fields. *American Sociological Review* Vol. 48(2), pp.147-160.

Parment, A., Wetterberg, G., Myren, V. (2009). *Med blicken framåt: Några framtida utmaningar och möjligheter för en attraktiv arbetsgivare*. Svergies Kommuner och Landsting (Ed). Stockholm: SKL Kommentus, pp. 3-137.

Parmar, A., (2014). The role of Hr Department in Employer Branding at Public and Private Sector. *Journal of Human Resources Management and Labour Studies* Vol. 2(2), pp.201-225.

Rampl, L. V., & Kenning, P. (2014). Employer brand trust and affect: linking brand personality to employer brand attractiveness. *European Journal of Marketing* Vol. 48(1/2), pp.218-236.

Robert, K. (2014). "Lovemarks in the Age of Now", in Kompella (ed), *The Definitive Book of Branding*. New Delhi, India: Sage Publications Inc. pp.369-393.

Röhmer, M. (2011). "Start varumärke? Ja vi har ju vår slogan...", in Komlitt (ed), *Talanger sätter ribban – vinnande personalstrategi för offentliga sektor*. Höganäs: Komlitt AB, pp. 46- 50.

Røvik, K.A. (2008). Management-Samhället Trender och idéer på 2000-talet. Malmö:Liber.

Sivertzen, A-M., Nilsen, E.R., Olafsen, A.H. (2013). Employer branding: employer attractiveness and the use of social media. *Journal of Product & Brand Management* Vol. 22(7), pp.473-483.

SKL; Sveriges Kommuner och Landsting. (2006). *Den lokala offentligheten- En studie av makt, medier och mötesplatser i 14 kommuner*. Retrieved February 25, 2015 from: <u>http://webbutik.skl.se/bilder/artiklar/pdf/7164-201-1.pdf</u>

SKL; Sveriges Kommuner och Landsting (2010). *Arbetsgivarperspektiv på kommuner och Landsting*. Västerås: Edita. Retrieved February 3, 2015 from: http://webbutik.skl.se/bilder/artiklar/pdf/7164-543-2.pdf SKL;Sveriges Kommuner och Landsting (2014). Vem jobbar i kommuner och landsting?. Retrieved 29 March, 2015 from:

http://www.sverigesviktigastejobb.se/fakta/huruppfattasvalfardsjobben/vemjobbarikommuneroch landsting.1999.html

SKL;Sveriges Kommuner och Landsting (2015). Sveriges Viktigaste Jobb Finns I Välfärden: Hur möter vi rekryteringsutmaningen? Retrieved April 7, 2015 from: file:///Users/josefinekoskinen/Downloads/7585-216-4.pdf

Sveriges Viktigaste Jobb. (2014). Om Sveriges Viktigaste Jobb. Retrieved April 7, 2015 from <u>http://www.sverigesviktigastejobb.se/omsverigesviktigastejobb.67.html</u>

Van Mossevelde, C. (2012). The Importance of Internal Employer Branding: The inside out approach. Retrieved March 11, 2015 from:

http://www.employerbrandingtoday.com/blog/2012/01/05/the-importance-of-internal-employerbranding/

Vergne, J.P.(2011). Toward a New Measure of Organizational Legitimacy: Method, Validation, and Illustration. *Organizational Research Methods* Vol 14(3), pp.484-502.

Wallace, M., Lings, I., Cameron, R., Sheldon, N.(n.d). *Attracting and Retaining Staff: The Role of Branding and Industry Image*. Retrieved February 18, 2015 from: http://link.springer.com/chapter/10.1007/978-981-4560-58-0\_2#page-1\_\_\_\_\_\_

Wellinder, E., Araujo, J., Lynn, K. (2011). *Talent management from an employer branding perspective*. Retrieved February 16, 2015 from: <u>http://www.employerbrandingtoday.com/blog/2011/04/11/talent-management-from-an-employer-branding-perspective/</u>

Wright, P. M., & McMahan, G.C (1992). Theoretical Perspectives for Strategic Human Resource Management. *Journal of Management* Vol. 18(2), pp. 295-320.

# 10. Appendix

## Appendix 1 - Interview guide

## **Background:**

1. What is your current position, and can you describe some of your daily work tasks?

2. How long have you worked within the Municipality of Gothenburg?

3. What are the goals of the organization? Does the organization have a vision?

4. Can you further explain the four work objectives that are common for how to work within the organization? Can you explain how these can bee seen in your daily work?

## **Employer branding:**

5. Can you explain what attractive employer means for you?

6. Why is it important to be seen as an attractive employer? What is the goal and purpose of having a strategy regarding the attractive employer perspective?

7. How do you work to be seen as an attractive employer, both for current as well for prospective employees? Can you give an example on an incentive and what you contributed with?

8. What actors are involved in the employer branding work within the organization? How do you experience the collaboration between the involved actors?

9. What are some challenges and issues while working with employer branding and to work with the image of the organization as an attractive employer?

10. How much impact do you perceive media and the things they write about your organization has on your daily work?

11. How do you like working within the public sector?

12. What makes you to an attractive employer? What do you believe could enhance the chances to attract people to apply for a job within your organization?

## Appendix 2 - Letter of consent

#### Hej xxx,

Jag är en student på Göteborgs Universitet som studerar masterprogrammet inom Strategiskt Personalarbete. Jag är nu inne på sista året och skriver för tillfället på min masteruppsats. Uppsatsen handlar om "employer branding" (attraktiv arbetsgivare) och hur en offentlig verksamhet arbetar med arbetsgivarerbjudandet i sitt dagliga arbete. Vidare berör studien hur medarbetarna förstår företagets arbetsgivarerbjudande.

Syftet med studien är att skapa förståelse för hur en strategi, som syftar till att framstå som en attraktiv arbetsgivare, används inom den offentliga sektorn. För att kunna genomföra min studie behöver jag din hjälp. Därför skulle jag vara ytterst tacksam om du, som medarbetare inom Göteborgs Stad, vill medverka i en intervju som kommer att ta ca.30-45min. Din medverkan ger dig möjlighet att bidra med värdefull kunskap. Resultaten kommer enbart användas i forskningsändamål.

Ditt deltagande är frivilligt och du kan välja att avstå från att svara på vissa frågor, eller avbryta intervjun när som helst, ifall så önskas. Din integritet kommer att skyddas och ditt deltagande kommer behandlas konfidentiellt. Du och dina svar kommer vara anonyma. Om du vill hjälpa mig genom att delta i studien, var vänlig att återkom till mig så bestämmer vi en dag och tid som passar dig.

Din medverkan är värdefull för mitt resultat! Har du ytterligare några frågor eller funderingar så är du välkommen att kontakta mig eller min handledare för mer information. Tack på förhand!

Ser fram emot att höra från er!

Vänliga hälsningar, Josefine Koskinen